An intervention to support adherence to inhaled medication in adults with cystic fibrosis: the ACtiF research programme including RCT

Martin J Wildman,^{1,2} Alicia O'Cathain,^{2*} Daniel Hind,³ Chin Maguire,³ Madelynne A Arden,⁴ Marlene Hutchings,¹ Judy Bradley,⁵ Stephen J Walters,² Pauline Whelan,⁶ John Ainsworth,⁶ Paul Tappenden,² Iain Buchan,^{6,7} Rachel Elliott,⁸ Jon Nicholl,² Stuart Elborn,⁵ Susan Michie,⁹ Laura Mandefield,² Laura Sutton,² Zhe Hui Hoo,^{1,2} Sarah J Drabble,² Elizabeth Lumley,² Daniel Beever,³ Aline Navega Biz,² Anne Scott,³ Simon Waterhouse,³ Louisa Robinson,³ Mónica Hernández Alava² and Alessandro Sasso²

- ¹Sheffield Adult Cystic Fibrosis Centre, Northern General Hospital, Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK
- ²School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK ³Clinical Trials Research Unit, University of Sheffield, Sheffield, UK
- ⁴Centre for Behavioural Science and Applied Psychology, Sheffield Hallam University, Sheffield, UK
- ⁵Wellcome-Wolfson Institute for Experimental Medicine, School of Medicine, Dentistry and Biomedical Sciences, Queen's University Belfast, Belfast, UK
- ⁶Health eResearch Centre, Division of Imaging, Informatics and Data Sciences, School of Health Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester Academic Health Science Centre, Manchester, UK
- ⁷Department of Public Health and Policy, Institute of Population Health Sciences, University of Liverpool, Liverpool, UK
- ⁸Division of Population Health, Health Services Research and Primary Care, School of Health Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK
- ⁹Centre for Behaviour Change, University College London, London, UK

*Corresponding author a.ocathain@sheffield.ac.uk

Declared competing interests of authors: Martin J Wildman reports funding from NHS England, PARI GmbH (Starnberg, Germany) and Koninklijke Philips N.V. (Amsterdam, the Netherlands) during the conduct of the study. His team received equipment and support for independent statistical advice to investigate the I-neb device (Koninklijke Philips N.V.) as a mechanism of detecting exacerbations. He received travel expenses to meet PARI GmbH in Munich to explain the CFHealthHub programme. His team has received funding for independent statistical advice to investigate the use of spirometric devices to understand lung function in the community and devices to support this study. The ACT if (Development and evaluation of an intervention to support Adherence to treatment in adults with Cystic Fibrosis) programme was supported by the NHS England Commissioning for Quality and Innovation (CQUIN) payment framework [funding identifier IM2 Cystic Fibrosis Patient Adherence (Adult)]. Alicia O'Cathain was a member of a subpanel for the National Institute for Health Research (NIHR) Programme Grants for Applied Research (PGfAR) programme (2007–17). Daniel Hind reports grants from the NIHR (10/57/46, 12/28/05, 12/144/04, 13/24/03, 15/178/09, 17/17/02, 17/72/02, NIHR127454 and 17/136/10), NHS England and Koninklijke Philips N.V. during the conduct of the study outside the submitted work. He was a member of the following committees during the project: NIHR Research for Patient Benefit, Yorks and North East Regional Advisory Committee (2016–20) and NIHR Health Technology Assessment (HTA) Clinical Evaluation and Trials Funding Committee (2019–20). Stephen J Walters reports grants from the Department of Health and Social Care, NIHR, Medical Research Council (MRC) and National Institute for Health and Care Excellence (NICE); personal fees from book royalties; personal fees from external examining; and an NIHR Senior Investigator award, outside the submitted work. He was a member of the following during the project: NIHR HTA Clinical Trials and Evaluation Committee (2011–17), NIHR HTA Commissioning Strategy Group (2012–17), NIHR PGfAR Committee (2020) and NIHR Pre-doctoral Fellowship Selection Committee (2019–20). Pauline Whelan reports that the University of Manchester software team received funds from PARI GmbH outside the submitted work. She is a director of Affigo C.I.C. (Altrincham, UK), a Manchester-based community interest company, which comprises a team of clinicians, academics and software engineers who work closely with mental health service users to develop innovative digital technologies. Affigo C.I.C. has not been involved in this project or publication. Pauline Whelan is the owner/director of Prism Life Ltd (Leeds, UK), a small research and software consultancy company. Prism Life Ltd has had no involvement in this work. John Ainsworth reports funding from PARI GmbH outside the submitted work. He is director of Affigo C.I.C. Iain Buchan reports grants from the MRC during the conduct of the study, personal fees and other funding from AstraZeneca plc (Cambridge, UK), and personal fees and other funding from Microsoft Corporation (Redmond, WA, USA), outside the submitted work. Stuart Elborn reports grants from the Innovative Medicines Initiative, Seventh Framework Programme, European Commission, the MRC and the United Kingdom Bronchiectasis Registry (BRONCH-UK) during the conduct of the study. Zhe Hui Hoo reports other funding from NHS England and PARI GmbH during the conduct of the study.

Published October 2021 DOI: 10.3310/pgfar09110

Plain English summary

ACtiF research programme including RCT Programme Grants for Applied Research 2021; Vol. 9: No. 11 DOI: 10.3310/pgfar09110

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

People with cystic fibrosis need medication to stay well. They take some medication using a nebuliser. This is a machine that helps people to inhale medicine as a mist. Nebulisers are used to take medication that prevents health problems occurring. People do not always take all the doses of their nebuliser treatments and they often believe that they take more than they do. Our research aimed to develop a new intervention called 'CFHealthHub'. This was a website and mobile application to show people with cystic fibrosis their actual medication use, and training for health-care professionals to help people to take more of their medication.

We used feedback from patients and staff in three cystic fibrosis centres to develop CFHealthHub. We tested it in a small study. We then improved the intervention before using it in a larger study. We focused on whether or not people with cystic fibrosis had fewer spells of being ill in hospital and whether or not they took more of their treatments.

A total of 19 cystic fibrosis centres took part in the main study. The study recruited 608 people with cystic fibrosis. They were in the study for 12 months. All participants received their usual treatment. Half were also given the CFHealthHub intervention. Numbers of nebuliser treatments taken by people with cystic fibrosis were automatically measured. People with cystic fibrosis completed questionnaires about their general health during the study. We interviewed 22 people with cystic fibrosis and 26 health-care professionals delivering the intervention about their views of CFHealthHub.

We found that CFHealthHub did not reduce spells of being ill in hospital. However, we found that those who used CFHealthHub took more of their nebuliser medication. People with cystic fibrosis felt that the new intervention was helpful and valued the relationship they had with health-care professionals delivering the intervention. We found that the intervention may be good value for money for the NHS. However, this depends on how long the intervention is effective for and whether or not increasing adherence also increases the numbers of prescriptions for medicines.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1212-20015. The contractual start date was in March 2015. The final report began editorial review in February 2020 and was accepted for publication in December 2020. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGFAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGFAR programme or the Department of Health and Social Care.

Copyright © 2021 Wildman *et al.* This work was produced by Wildman *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk